Author's response to reviews

Title: Acupuncture for patients with functional dyspepsia: design and methods of a randomized controlled trial

Authors:

Hui Zheng (zhhui126@yahoo.com.cn)
Xiao-ping Tian (ttxp2006@126.com)
Ying Li (jialee@mail.sc.cninfo.net)
Fan-rong Liang (acuresearch@126.com)

Version: 3 Date: 6 July 2009

Author's response to reviews: see over
Dear editors,
The reviewer has made very good suggestions here. In order to improve the language quality, we asked a teacher who majors in English to edit the manuscript for us. Changes to this manuscript were highlighted using track changes mode in MS word.
If this revised manuscript still needs further editing, please kindly point out where the problem is, and we will keep improving.

Sincerely yours,
Fan-rong Liang
Tel.: 028- 66704320
Fax: 86-28-87784606
E-mail: acuresearch@126.com